

EO-1 Weekly Report


	Earth Observing – 1

	June 13, 2012 – June 19, 2012

	DOY
	165
	–
	171

	Mission Day
	4283
	–
	4289


EO-1 SPACECRAFT

Instruments

· Scheduled 117 science Data Collection Events (DCEs) this week 

· Four images not taken see the discrepancy section for more information
· Performed instrument decontamination cycles

· Conducted HSI deicing from 170/00:25z to 170/15:15z

· Conducted ALI outgassing from 170/00:35z to 170/15:25z

· Performed Instrument calibration

· Solar Array Characterization at 170/10:41z

· HSI solar calibration 170/20:33z

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintaining nominal spacecraft state of health

· Creating a script to check if the ASE goal for next day is loaded

· Adding to the script the functionality that the end-to-end_check script currently has, checking if the science quicklook was downlinked on the last pass

· Adding a part to display what station and time the next 3 passes are on

· Preparing to upload X-band target table containing TrollSat
· Creating scripts to transfer primary ASIST T&C and update the central file hub that we are building.

· Adding functionality to prepare the target ASIST system for primary use automatically

· Researching all files received on ASIST systems to see if they can all be sent to the central file hub and then transferred all Realtime systems

Mission Planning
· Working with White Sands for possible contacts replacements to MG1
· Using MG1 as best effort

· Working with JPL to configure Lunar Calibration sequences 

· Researching updates on MP files to determine if they can be gathered on the central file hub and then transferred to both MP systems
Flight Dynamics
· Experimenting with various lunar calibration timings and instrument configuration
· Designed and created single, slow scan, Hyperion only Lunar Image

· Slight changes need to be made to the sequence

· The first attempt showed the moon coming into view about 20 seconds into the image, moving the start time of the image to be ¼ of the image duration before the lunar crossing time, the first attempt has ½ of the image before the crossing time.
· Working with ASPEN technicians to rebuild lunar calibration

· Gave the ASPEN technicians the output from flight dynamics that would occur with a 4 scan lunar calibration (ALI and HYP) for design of the new sequence

· Talk about whether it will be dual 3-scan calibrations or a single ALI only 4 scan and a Hyperion only single scan

· Researching updates to FD files to determine if they can be gathered on the central file hub and then transferred to both FD systems
· Checking TR2 masking file for elevation angle for EO-1 passes from 20:00z-22:00z

· Determining if there are any other areas where we lose lock

System Administration

· Completed installation of a new power supply in the flight software lab development computer. The system now boots nominally.

· Supporting upcoming ESMO ITAP audit.

· Updated the EO-1 network inventory spreadsheet and sent it to IT security personnel.

· Sent a copy of the most recent quarterly IONet scan results to IT security personnel.

· Attended the kickoff/overview meeting and the EO-1 subsystem discovery meeting with the audit team.

· Completed annual CIS Benchmark review on the Flight Software Lab ASIST T&C system. Verified previously applied benchmark settings and implemented new benchmark items. This review was documented in a benchmark spreadsheet.

· Completed applying the June 2012 updates/patches on the CNE computers. The patches included Adobe Flash, Firefox, Java and Microsoft products.

· Reported May POAM status update to IT Security team.

· Began taking the required “Backup and Security Settings in Microsoft Windows XP” course on SATERN.

· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, and MOC CNE computer.

· Created weekly tape.

GROUND AND SPACE NETWORK

Station Downtimes 

· SGS has had an equipment malfunction and latest indication is that it will be back up August 2012
· HGS is down, unknown time to be back up

· MG1 is back online after the electrical problems
Operational Discrepancies

· Loosing telemetry on all MG1 passes prior to DOY167 due to station electrical problems
· MG1 is back online and has successful passes

· Investigating why the spacecraft did not take 4 images on DOY 166. 
· The WARP did not play anything back because nothing was recorded. 

· This problem had similar circumstances as the previous problem on 162. 
· The WARP technician is performing tests on the on ground ASE system to see if we can recreate the problem and determine the cause.

Number of Real-Time Supports

[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

13-Jun-12 165 4 8 1 0 13

14-Jun-12 166 6 3 0 0 9

15-Jun-12 167 5 7 0 0 12

16-Jun-12 168 3 4 1 0 8

17-Jun-12 169 6 3 0 0 9

18-Jun-12 170 2 8 0 0 10

19-Jun-12 171 6 3 0 0 9

32 36 2 0 70 Weekly Totals


UPCOMING EVENTS

· X-band target table upload

· Redesign of nominal Lunar Calibration
· Adjustments to the design of the Single, slow, Hyperion lunar image

· TR2 X-Band pass testing
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		13-Jun-12		165		4		8		1		0		13

		14-Jun-12		166		6		3		0		0		9

		15-Jun-12		167		5		7		0		0		12

		16-Jun-12		168		3		4		1		0		8

		17-Jun-12		169		6		3		0		0		9

		18-Jun-12		170		2		8		0		0		10

		19-Jun-12		171		6		3		0		0		9

		Weekly Totals				32		36		2		0		70
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